Author's response to reviews

Title: Undocumented migrants lack access to pregnancy care and prevention

Authors:

Hans Wolff (hans.wolff@hcuge.ch)
Manuella Epiney (manuella.epiney@hcuge.ch)
Ana P Lourenco (lourenco-ana-paula@diogenes.hcuge.ch)
Michael C Costanza (michael.costanza@hcuge.ch)
Jacqueline Delieutraz-Marchand (delieutraz-marchand-jacqueline@diogenes.hcuge.ch)
Nicole Andreoli (Nicole.Andreoli@hcuge.ch)
Jean-Bernard Dubuisson (dubuisson-jean-bernard@diogenes.hcuge.ch)
Jean-Michel Gaspoz (gaspoz-jean-michel@diogenes.hcuge.ch)
Olivier Irion (Olivier.Irion@hcuge.ch)

Version: 9 Date: 18 March 2008

Author's response to reviews:

The following changes were made:

Table order - The first table referenced in the manuscript text should be Table 1, the second, Table 2, etc. The tables should be provided in this same order at the end of the manuscript.

Please move the Competing interests to above the authors' contributions